**Author’s response to reviews**

**Title:** Case Report: A fatal case of disseminated adenovirus infection in a non-transplant adult haematology patient

**Authors:**

Michael Joffe (Michael.C.Joffe@uhl-tr.nhs.uk)

Simon Wagner (Simon.Wagner@uhl-tr.nhs.uk)

Julian Wei Tze Julian Wei Tze (jwtang49@hotmail.com;jwtang49@gmail.com)

**Version: 3 Date: 08 Jan 2018**

**Author’s response to reviews:**

INFD-D-17-00238R2

Case Report: A fatal case of disseminated adenovirus infection in a non-transplant adult haematology patient

Michael C Joffe, MBChB; Simon Wagner, MBChB MD PhD FRCP; Julian Wei Tze Tang, BA MBChB MA PhD MRCP FRCPath FHKCP FHKAM

BMC Infectious Diseases

**Editor Comments:**

Many thanks for attaching the Consent for publication obtained from the wife of the patient. This should now be removed, alongside the other additional files.

- These files have been deleted, as requested.

In addition to removing the additional files, please also revise the following:

1) Ethics approval and consent to participate: please revise this to clarify whether ethical approval and consent to participate (i.e., be treated) were obtained. If the patient was treated according to standard care, then you may write "not applicable" here and clarify that the patient was treated according to standard care.

- This has been amended, as requested.
2) Consent to publish: please include this heading after Ethics approval and consent to participate, and state here that consent for the publication of the case report and any accompanying images was obtained from the patient's next of kin.

- This has been amended, as requested.